Author's response to reviews

Title: Intraoperative parathyroid hormone assay during focused parathyroidectomy: the importance of 20 minutes measurement

Authors:

Pietro Giorgio CALO' (pgcalo@unica.it)
Giuseppe Pisano (gpisano@unica.it)
Giulia Loi (giulialoi@hotmail.it)
Fabio Medas (fabiomedas@yahoo.it)
Lucia Barca (lucia.barca@gmail.com)
Matteo Atzeni (teo.atzeni@tiscali.it)
Angelo Nicolosi (nicolosi@unica.it)

Version: 2 Date: 13 April 2013

Author's response to reviews:

We submit the revised version of our article entitled "Intraoperative parathyroid hormone assay during focused parathyroidectomy: the importance of 20 minutes measurement" for the publication in your journal. We performed all the correction requested:

1) We specified the individual contributions of authors to the manuscript.
2) We declared no competing interests.
3) We add an acknowledgements section and we specified to have no acknowledgement.
4) We confirm that ethical approval for our study was obtained from our institutional ethical committee and include a statement concerning this in our manuscript.
5) We confirm informed consent was obtained from patients for their inclusion in our study and include a statement concerning this in our manuscript.
6) We confirm that this is a retrospective study and we clearly stated it in our manuscript.

All persons listed as authors have given their approval for the submission of the paper.

Pietro Giorgio Calò
Giuseppe Pisano
Giulia Loi
Fabio Medas
Lucia Barca
Matteo Atzeni